\documentclass[11pt]{article}
%\usepackage{amsmath}
%\usepackage{amsfonts}
%\usepackage{amssymb}
\usepackage[letterpaper, margin=1in]{geometry}
%\usepackage{graphicx}
\usepackage{hyperref}
%\usepackage{fullpage}

\begin{document}

\begin{flushleft}
\begin{Large}
    \textbf{Name: \underline{\hspace{4in}}}
\end{Large}    

\bigskip

\begin{center}
    \begin{large}
        \textbf{\underline{Letter for Assignment 4 Final Submission}}
    \end{large}
\end{center}

\bigskip


\textbf{Directions:} Please use the following text as a template for a letter addressed to me for the final submission of your  
\emph{Cover Letter and Resume Writing Assignment 4}. In addition to submitting the \texttt{.pdf}, \texttt{.tex} and other files for your assignment in the \texttt{assignment4final} folder, also include the 
 \texttt{.tex} file you create by writing a letter to me using the template 
 (\texttt{\href{https://egunawan.github.io/writing/hw/04/assignment04finalcover.tex}{assignment04finalcover.tex}})
 that contains the following prompts. 

\vspace{.25in}

Dear Professor,

\bigskip

In this assignment, I am trying to...

\bigskip

In my first submission, I...

\bigskip

(Optional) Some of the feedback that I received from the UConn Career Center or Writing Center on my first submission were ...

Some of the feedback that I received from my peers (classmates) in class on my first submission were ...

Given the feedback I received from from my peers (and the Career Center or Writing Center, if applicable), I decided to...because...

\bigskip

Given the feedback I received from you on my first draft, I decided to...because...

\bigskip

For this final submission, I concentrated most of my efforts on...because ...

\bigskip

I think that the strongest parts of my assignment are... 

\bigskip

I am especially proud of...

\bigskip

What I struggled with the most was...

\bigskip

Some things you may want to keep in mind as you read this assignment are ...

\bigskip

Sincerely,

\bigskip

Your Name

\end{flushleft}

\end{document}

